Author's response to reviews

Title: Classic Yin and Yang Tonic Formula for osteopenia : Study protocol for a randomized controlled trial

Authors:

Feng Yang (yangfengdudu@163.com)
De-zhi Tang (dztang702@hotmail.com)
Xue-jun Cui (diyicuixi@163.com)
Jonathan D. Holz (jonathan_holz@urmc.rochester.edu)
Qin Bian (bianqin213@126.com)
Qi Shi (shigish@hotmail.com)
Yong-jun Wang (yjwang88@hotmail.com)

Version: 7 Date: 25 July 2011

Author's response to reviews: see over
Dear Editors and reviewers:

Thank you for your hard work. We would like to resubmit our manuscript “Classic Yin and Yang Tonic Formula for osteopenia : study protocol for a randomized controlled trial ” (MS: 1725176005228941) to the Trials for consideration for publication. We have cited figure 1 in the text and corrected typographical errors marked with blue.

Sincerely,

Yang Feng

Spine Research Institute,
Shanghai University of Traditional Chinese Medicine,
725 South Wanping Road,
Shanghai, 200032, China
Office: 86-21-6428-6239
Fax: 86-21-6428-6239